**Answering Reviewer**

**Specific Comments to Authors:** This is a case highlights the difficulty of establishing a definitive early diagnosis of anorectal melanoma, the importance of performing histological analysis on a well-represented biopsy specimen and the poor prognosis, even with radical surgery. This case reports unique findings and is theoretically based on the current literature. The subject is very important to clinical oncology practice. The paper is well written and provides useful information for the readers. This case is well presented. It makes important points which are widely applicable.

There are no specific requirements regarding the manuscript.